Author's response to reviews

Title: Efficacy of Vitamin D Supplementation in Multiple Sclerosis (EVIDIMS Trial): study protocol for a randomized controlled trial

Authors:

Jan Dörr (jan-markus.doerr@charite.de)
Stephanie Ohlraun (stephanie.ohlraun@charite.de)
Horst Skarabis (dr.skarabis@online.de)
Friedemann Paul (friedemann.paul@charite.de)

Version: 2 Date: 26 January 2012

Author's response to reviews:

Dear Editors of TRIALS,

please find attached the slightly revised manuscript in which we have made some minor format changes and have corrected the financial supporter’s name.

Yours sincerely

Stephanie Ohlraun - on behalf of the corresponding author Jan Doerr